Author's response to reviews

Title: A prospective, randomized, double-blinded single-site control study comparing blood loss prevention of tranexamic acid (TXA) to epsilon aminocaproic acid (EACA) for corrective spinal surgery

Authors:

Kushagra Verma MS (vermak01@med.nyu.edu)
Thomas J Errico MD (Thomas.Errico@nyumc.org)
Kenneth M Vaz BS (kenneth.vaz@gmail.com)
Baron S Lonner MD (BLonner@nyc.rr.com)

Version: 5 Date: 11 March 2010

Author's response to reviews:

Dear Editors,

Major revisions have been made as requested.